Author’s response to reviews

Title: Inhibitory effects of berberine on proinflammatory M1 macrophage polarization through interfering with the interaction between TLR4 and MyD88

Authors:

Jing Gong (jgongtcm@126.com)
Jingbin Li (tjlaojing@163.com)
Hui Dong (tjhdonghui@163.com)
Guang Chen (75240882@qq.com)
Xin Qin (1396806232@qq.com)
Meilin Hu (497079049@qq.com)
Fen Yuan (630573650@qq.com)
Ke Fang (370440692@qq.com)
Dingkun Wang (657034873@qq.com)
Shujun Jiang (490520605@qq.com)
Yan Zhao (164119173@qq.com)
Wenya Huang (511031505@qq.com)
Zhaoyi Huang (253573064@qq.com)
Fuer Lu (jgongtcm@163.com)

Version: 3 Date: 25 Sep 2019

Author’s response to reviews:

Dear Editor Ismail Shah,

Thank you for your kind letter on September 25, 2019. We revised the manuscript in accordance with the reviewers’ and editors’ comments, and carefully proof-read the manuscript to minimize the errors. Here below is our description on revision according to the comments.
The reviewers have thoroughly checked the manuscript and have suggested to improve English and omit grammatical errors present in the manuscript.

The author’s answer: I am sorry for the grammar errors in the paper. We have already asked a native speaker on American Journal Experts to reedit our paper and proof-read the manuscript to minimize the errors carefully.

Thank Editor and all the reviewers for the kind advices.

Sincerely yours,

Fuer Lu